Author's response to reviews

Title: Whole body vibration compared to conventional physiotherapy in patients with gonarthrosis: a protocol for a randomized, controlled study

Authors:

Gregor Stein (gregor.stein@uk-koeln.de)
Peter Knoell (peter.knoell@uk-koeln.de)
Christoph Faymonville (christoph.faymonville@uk-koeln.de)
Thomas Kaulhausen (thomas.kaulhausen@uk-koeln.de)
Jan Siewe (jan.siewe@uk-koeln.de)
Christina Otto (christina.otto@uk-koeln.de)
Peer Eysel (peer.eysel@uk-koeln.de)
Kourosh Zarghooni (kourosh.zarghooni@uk-koeln.de)

Version: 4 Date: 20 June 2010

Author's response to reviews:

Dear ladies and gentlemen,

according to your e-mail received the 17th of june, I made the required changes to the manuscript:
- Authors list was updated and is now identically to the list in the submission system
- Trial registration is now also mentioned in the submission system
- A statement that all authors read and approved the final manuscript was inserted
- Reference citation has been changed to consecutive numbers in square brackets
- The table section has been removed, an appendices section has been inserted

All minor changes were also performed.

Kind regards,

Gregor Stein